Author's response to reviews

Title: Qualitative evaluation of a diabetes electronic decision support tool: views of users

Authors:

Qing Wan (q.wan@unsw.edu.au)
Meredith Makeham (m.makeham@unsw.edu.au)
Nicholas A Zwar (n.zwar@unsw.edu.au)
Susanna Petche (susannapetche@hotmail.com)

Version: 2 Date: 21 March 2012

Author's response to reviews:

The Editor,
BMC Medical Informatics and Clinical Decision Making,

As requested details of the human ethics approval and a statement of compliance with the Helsinks Declaration have been added at the start of the methods section.

It is stated in the methods section (under recruitment) that written informed consent for participation in the study was obtained from all participants.

Yours sincerely,
Nicholas Zwar